Author’s response to reviews

Title: Long-term effectiveness and costs of a brief self-management intervention in women with pregnancy-related low back pain after delivery

Authors:

Caroline HG Bastiaenen (chg.bastiaenen@epid.unimaas.nl)
Rob A de Bie (ra.debie@epid.unimaas.nl)
Johan WS Vlaeyen (j.vlaeyen@dep.unimaas.nl)
Marielle EJB Goossens (m.goossens@dep.unimaas.nl)
Pieter Leffers (p.leffers@epid.unimaas.nl)
Pieter JC Wolters (pj.wolters@hetnet.nl)
Janneke M Bastiaanssen (j.bastiaanssen@epid.unimaas.nl)
Piet A van den Brandt (pa.vandenbrandt@epid.unimaas.nl)
Gerard GM Essed (ges@sgyn.azm.nl)

Version: 4 Date: 29 May 2008

Author’s response to reviews: see over
Dear Sir,

Hereby, I send you the revised version of the paper: “Long-term effectiveness and costs of a brief self-management intervention in women with pregnancy-related low back pain after delivery” for publication in BMC Pregnancy and Childbirth.

The following changes to the format of the manuscript are made:

1. The authors’ qualifications (e.g. PhD/MD) are removed from the manuscript and submissions system,
2. Word count is removed,
3. The www.controlled-trials.com is removed from the trail registration,
4. The “Authors Contribution” is changed to “Authors’ contributions” and there is stated that all authors read and approved the final manuscript,
5. The section “Figures is renamed in “Figure legends”,
6. Website addresses are removed from the text,

Thank you

Kind regards,

Caroline Bastiaenen, PhD
Department of Epidemiology
Maastricht University
PO Box 616
6200 MD Maastricht
The Netherlands

e-mail: chg.bastiaenen@epid.unimaas.nl